Author’s response to reviews

Title: Rapid Development of HIV Elite Control in a Patient with Acute Infection

Authors:
Deirdre Morley (deemorley@gmail.com)
John Lambert (jlambert@mater.ie)
Louise Hogan (lou.hogan3@gmail.com)
Cillian De Gascun (cillian.degascun@ucd.ie)
Niamh Redmond (n.redmond@ucd.ie)
Rachel Rutishauser (Rachel.Rutishauser@ucsf.edu)
Cassandra Thanh (Cassandra.thanh@ucsf.edu)
Erica Gibson (gibsonerica14@gmail.com)
Kristen Hobbs (Kristen.Hobbs@ucsf.edu)
Sonia Bakkour (SBakkour@vitalant.org)
Michael Busch (MBusch@vitalant.org)
Jeremy Farrell (jfarrell@mater.ie)
Padraig McGetrick (padraigmcgetrick@gmail.com)
Timothy Henrich (Timothy.Henrich@ucsf.edu)

Version: 1 Date: 04 Jun 2019

Author’s response to reviews:

The consent to publish form provided by the Editor has been signed by the patient. The patient is aware that anonymity cannot be fully guaranteed and that there is a possibility that they could be identified based on the case report information/images.

A copy of this consent form is available on request.